Author’s response to reviews

Title: Electroencephalographic Evaluation of Acoustic Therapies for the Treatment of Chronic and Refractory Tinnitus

Authors:
Luz Maria Alonso-Valerdi (lm.aloval@itesm.mx)
David Ibarra-Zarate (david.ibarra@itesm.mx)
Francisco Tavira-Sánchez (franj.tavira@gmail.com)
Ricardo Ramirez-Mendoza (ricardo.ramirez@itesm.mx)
Manuel Recuero (manuel.recuero@upm.es)

Version: 2 Date: 10 Nov 2017

Author’s response to reviews:

Dear Reviewers and Editor,

Concerning your issues, we have amended the manuscript as follows:

1. In the trial registration statement of your abstract, please include the name of the registry where your trial was registered. Currently, we cannot find the trial registration page with the given registration number.

The registration trial information has been updated as

Registration Number: ISRCTN14553550
ISRCTN Registry: BioMed Central
Date of Registration: October 31st, 2017

2. Please note that we recommend using a repository that gives permanent links rather than Google drive links, as these are not always permanent. The questionnaire can also be uploaded as a supplementary file.

Questionnaires and supplementary materials have been uploaded in "figshare". The corresponding URLs have been added in text as a citation as was required in the point 3.
3. Please reformat your References and in-text citations according to our submission guidelines:

Citations have been reformatted according to the numerical method.

4. Upon uploading your revisions, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript.

A clean manuscript has been attached.

Thanks for your feedback and we're looking forward to hearing from you.

All the best

Authors